Author’s response to reviews

Title: COULD CURRENT FACTORS BE ASSOCIATED WITH RETROSPECTIVE SPORTS INJURIES IN BRAZILIAN JIU-JITSU? A CROSS-SECTIONAL STUDY

Authors:

Dayana das Graças (dayana.dg.fisio@gmail.com)

Letícia Nakamura (leticiahah@hotmail.com)

Fernando Sérgio Silva Barbosa (fernandossb@outlook.com)

Paula Felippe Martinez (paulafmartinez@yahoo.com.br)

Filipe Abdalla Reis (filipe@byofisio.com.br)

Silvio Assis de Oliveira-Junior (oliveirajr.ufms@gmail.com)

Version: 3 Date: 22 Jun 2017

Author’s response to reviews:

Jodie Louise Wilkie, Ph.DAssociate Editor - Training, injury prevention, and biomechanicsBMC Sports Science, Medicine and RehabilitationE-mail: bmcsporsscimedrehabil@biomedcentral.comDear Dr. Jodie Louise,Thank you very much for the opportunity to resubmit the present manuscript (SSMR-D-16-00062R1). Actually, the answer and last adjustment are addressed in the enclosed response to Reviewer 1. We appreciate your careful attention and believe this revised manuscript is substantially improved due to offered suggestions. Thank your very much for your attention.Yours sincerely,Silvio Assis Oliveira Junior, Ph.DAddress for CorrespondenceFederal University of Mato Grosso do Sul, Integrated School Clinic Campo Grande, MS, Brazil. 79070-900. Campo Grande, MS, Brazil. Phone: 55-67-3345-7837 E-mail address: oliveirajr.ufms@gmail.br (S.A.Oliveira-Junior)Reviewer reports:Reviewer's Responses to Questions: Leonardo Vidal Andreato (Reviewer 2)1) General Comments: The authors answered all my questions. The text has been significantly improved. I do not have any more suggestions. Only, adjust the legend (symbols) of the Table 1. Response: Thank you very much for your careful review of the manuscript. We appreciated your suggestions and included some new information in the Table 1 as requested.